Author’s response to reviews

Title: Prevalence of Dental Caries and Associated Factors among 12 Years Old Students in Eritrea

Authors:

Amanuel Andegiorgish (akidane2016@gmail.com)
Bizen Weldemariam (wellzen2014@gmail.com)
Meron Kifle (meronmehari121@gmail.com)
Filmon Mebrahtu (filmongeb8@gmail.com)
Henos Zewde (henoskef95@gmail.com)
Micheal Tewelde (mikongre96@gmail.com)
Mohammed Hussen (hurybadre@gmail.com)
Winta Tsegay (wintakes95@gmail.com)

Version: 5 Date: 12 Dec 2017

Author’s response to reviews:

1) Declarations Heading

Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements
Reply: We have now added “Declarations Heading” for this section

2) Consent to Participate

Please confirm whether informed written consent was obtained from the parent or legal guardian of each child that participated in this study. Permission from the school is not sufficient to include minors in a study without parental consent.

Reply: We have now modified this section as” Ethical clearance …. and from the parent or legal guardian of each child that participated in this study……..

3) Author Information

Please remove the author information section from the declarations.

Reply: We have now removed this section

4) Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We have now made the finale clean format. Additional files are also put as separate files.